Author's response to reviews

Title: Concomitant septic arthritis and osteomyelitis of the hip in young children; a new pathophysiological hypothesis suggested by MRI enhancement pattern

Authors:

Laura Merlini (laura.merlini@hcuge.ch)
Mehrk Anooshiravani (mehrk.dumont@hcuge.ch)
Dimitri Ceroni (dimitri.ceroni@hcuge.ch)

Version: 5 Date: 16 April 2015

Author's response to reviews: see over
Dear Editor,

Thank you for your comments.

As you requested,

1. We refer to each author individually within the "contribution of authors" section (Page 10) (Page 13).

2. We changed "Patients and methods" on "Methods" both within the Abstract and the manuscript (Page 11 and Page 4).

3. We removed the duplicate figure.

Best regards

Laura Merlini